**Author's response to reviews**

**Title:** Refractile superficial retinal crystals and chronic retinal detachment: Case Report

**Authors:**

Maged S Habib (mhabib jc@yahoo.co.uk)
Sinead Byrne (sineadacbyrne@hotmail.com)
John H McCarthy (john.mccarthy@glos.nhs.uk)
David H W Steel (David.Steel@chs.northy.nhs.uk)

**Version:** 5 **Date:** 25 December 2005

**Author’s response to reviews:**

Dear Sir

Here we are submitting a revised version for the paper "Refractile superficial retinal crystals and chronic retinal detachment: Case Report".

We have corrected the requested points
1. The authors qualifications were removed from first page
2. The title and the authors list was removed from the second page
3. We added the competing interest section
4. The references list were completed with all the authors names cited.
5. The legends were removed from the figures files and added at the end of the main manuscript.

We have uploaded the revised version,

Kind regards

Maged Habib